

G OPEN ACCESS

Citation: Murenjekwa W, Chidhanguro K, Cowan FM, Lampe FC, Johnson C, Mpofu A, et al. (2025) Who has never tested for HIV following a community-based distribution of HIV self-test kits? Establishing associated predictors in rural Zimbabwe. PLOS Glob Public Health 5(4): e0004459. https://doi. org/10.1371/journal.pgph.0004459

Editor: Julia Robinson, PLOS: Public Library of Science, UNITED STATES OF AMERICA

Received: August 9, 2024

Accepted: March 3, 2025

Published: April 30, 2025

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pgph.0004459

Copyright: © 2025 Murenjekwa et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

RESEARCH ARTICLE

Who has never tested for HIV following a community-based distribution of HIV self-test kits? Establishing associated predictors in rural Zimbabwe

Wellington Murenjekwa^{1,2*}, Kudzai Chidhanguro¹, Frances M. Cowan^{1,3}, Fiona C. Lampe⁴, Cheryl Johnson⁵, Amon Mpofu⁶, Getrude Ncube², Owen Mugurungi², Karin Hatzold⁷, Elizabeth L. Corbett^{8,9}, Andrew N. Phillips⁴, Euphemia Sibanda^{1,3}, Valentina Cambiano⁴

 CeSHHAR Zimbabwe, Harare, Zimbabwe, 2 AIDS and TB Unit, Ministry of Health and Child Care, Harare, Zimbabwe, 3 Department of International Public Health, Liverpool School of Tropical Medicine, Liverpool, United Kingdom, 4 Institute for Global Health, University College London, London, United Kingdom, 5 Global HIV, Hepatitis and STI Programmes, World Health Organisation, Geneva, Switzerland, 6 National AIDS Council, Harare, Zimbabwe, 7 Population Services International, Cape Town, South Africa, 8 Department of Clinical Research, London School of Hygiene & Tropical Medicine, London, United Kingdom, 9 TB-HIV Group, Malawi-Liverpool-Wellcome Trust Clinical Research Programme, Blantyre, Malawi

* wellington.murenjekwa@ceshhar.org

Abstract

In 2023, Zimbabwe attained the 95-95-95 UNAIDS targets. However, some subpopulations are substantially less likely to have tested for HIV. Knowledge of characteristics of these groups is crucial in designing interventions that address their needs. We estimated the prevalence and predictors of "never-having tested for HIV" status following community-based distribution of HIV self-test kits in rural Zimbabwe. We analysed data from a household survey conducted as part of a cluster randomised trial comparing two community-based HIVST distribution models in six rural districts in 2018-19. HIVST distribution was conducted over one month, followed by the household survey after four months. Survey participants aged 16 years and above completed self-administered Audio-Computer-Assisted-Survey-Instrument. Unadjusted and adjusted mixed effect logistic regression was used to identify factors associated with never-having-tested for HIV. Of the 11,076 analysed participants, the median (IQR) age was 32(22,45) years and 54.5% were female. Seventeen percent of participants had never tested for HIV, primarily due to a perceived lack of HIV risk (50%). Never testers were more likely to be: men (adjusted odds ratio [AOR]=1.69;95%Confidence Interval [CI]=1.52-1.87); younger (16-24 years (AOR=3.84; 95%CI=3.23-4.55), 25-34 years (AOR=1.30; 95%CI=1.07-1.59)) and at-least 45 years old: (AOR=2.17; 95%CI=1.80-2.60); having lower levels of education: primary/less (AOR=1.68; 95%CI=1.46-1.98), some secondary (AOR=1.62; 95%CI=1.42-1.86) compared to at least complete secondary, unemployed (AOR=1.39; 95%CI=1.15-1.69); never married (AOR=3.48; 95%CI=2.98-4.07) and previously married (AOR=1.41; 95%CI=1.19-1.68) compared to currently married; having stigmatizing beliefs **Data availability statement:** Data was uploaded as Supporting information.

Funding: This work was supported by the Medical Research Council (grant number MR/ T042796/1); UNITAID STAR project (PO# 10140-0-600 and PO# 8477-0-600 to WM). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

(AOR=1.42; 95%CI=1.24-1.62); having: low (AOR=1.52, 95%CI=1.32-1.74) and medium (OR=1.53, 95%CI=1.33-1.75) levels of treatment optimism; not participating in household decisions (AOR=1.96; 95%CI=1.70-2.27) and not reporting condomless sex (AOR=2.58; 95%CI=2.31-2.87). The Ministry of Health need to scale up acceptable and targeted interventions to improve HIV testing in different subpopulations which includes but not limited to young people, unmarried, unemployed, those with stigmatizing beliefs and those not participating in decision making.

Author summary

Despite the increase in number of HIV testing approaches and of HIV tests, we still have many people who have never tested for HIV in their lifetime. Knowing the characteristics of people who have never tested for HIV, even after free distribution of HIV self-testing kits in their communities, is important to develop testing strategies that address their needs. We, therefore, estimated the prevalence of never having tested for HIV in these rural communities and identified the associated characteristics. We found that about one in five had never tested for HIV and half of these reported "no HIV risk perception" as the reason for not testing. Characteristics associated with not testing were being young, having a low level of education, being unemployed, unmarried, lack of involvement in decision making, having stigmatizing beliefs, not believing that antiviral therapy (ART) can improve health outcomes among people living with HIV and having sex using condom. There is need to offer acceptable and population specific interventions, to improve HIV testing in different subpopulations.

Introduction

HIV has claimed many lives globally and an estimated 38.4 million people were living with HIV by end of 2021 [1,2]. Globally, the burden of HIV is highest within the Eastern and Southern Africa (ESA) region which accounts for more than half (54%) of all people living with HIV (PLHIV) in the world [3]; while accounting for 7% of the world population [4].

Although the estimated numbers of new HIV infections has declined by 44% and estimated AIDS related deaths by 58% between 2010 and 2021in ESA compared with other regions [3], more needs to be done if the region is to achieve the goal of ending AIDS as a public health threat by 2030 [5]. In Zimbabwe, new HIV infections decreased by an estimated 40% between 2015 and 2021 [3] and HIV prevalence among adults 15-49 years fell from 18.1% in 2005 [6] to 11.8% in 2020 [7]. The reduction in HIV incidence and prevalence was mainly due to reduction in sexual risk behaviour [8,9], increase in the number of people living with HIV on antiretrovirals [10], and implementation of the Extended Zimbabwe National HIV and AIDS Strategic Plan (ZNASP III) 2015-2020 [11], which aimed to achieve the 90-90-90 targets by 2020 and focused on high-impact interventions for key and vulnerable populations. Key program activities included combination HIV prevention strategies, social and behaviour change communication campaigns, widespread condom promotion and distribution, the promotion of male circumcision and robust prevention of mother-to-child transmission programs.

Despite the country achieving 95-95-95 [12,13] UNAIDS 2025 targets (95% of people living with HIV knowing their status, 95% of those who know their status receiving treatment and 95% of those on treatment being virally suppressed), there are specific subpopulations, such

as men and key populations (female sex workers and men who had sex with men (MSM)) and especially young people (including children), that are still lagging behind [14–16]. These inequalities are probably due to lack of innovative case finding in children [16], offer of HIV testing being primarily in clinics and discriminating laws against MSM and sex workers [14].

HIV testing is the gateway to HIV prevention, care, and treatment services. In 2014 the Ministry of Health and Child Care (MOHCC) highlighted the need for increased access to HIV testing services, given the compelling HIV infection rates in the country [17]. In 2016 MOHCC then recommended HIV self-testing (HIVST) as an additional approach to HIV testing services [18] following extensive STAR consortium work on HIVST. Although willingness and interest to use HIVST was quite high, access to and awareness of HIVST was low in Zimbabwe [19].

The annual number of conventional HIV tests performed in Zimbabwe increased from 1.7 million in 2011 to 3 million in 2018 [20] and, as a consequence, the percentage of adults (15-49 years old) who were ever tested and had received results for HIV: among men from 16.4% in 2005 to 76.3% in 2020 and among women from 21.7% in 2005 [21] to 84.9% in 2020 [7]. This increase in number of HIV tests is the result of different testing approaches introduced by the MOHCC: provider-initiated testing and client-initiated testing, HIVST (facility- and community-based) with those with reactive results being confirmed with conventional testing, and index testing. Despite the expansion in HIV testing approaches and increase in number of tests, there is still a considerable number of people who have never tested for HIV in their lifetime. Knowledge of characteristics of people who have never tested for HIV, particularly in the context of community-based promotion of HIV testing, is crucial for developing testing models that target those specific subpopulations. Therefore, this study estimated the prevalence of never having tested for HIV among people living in rural areas of Zimbabwe and identified the associated characteristics. We also assessed factors associated with reporting no perceived HIV risk among participants who never tested for HIV.

Methods

Ethics statement

The research used anonymized secondary data from a study conducted by the author team, previously approved by Medical Research Council of Zimbabwe (MRCZ/A/2323), London School of Hygiene & Tropical Medicine Ethics Committee (15801) and WHO Ethical Review Committee (ERC.0003065); and informed written consent was obtained from all the participants. Prior to conducting the research, verbal consent to engage with the communities was obtained from local community leaders (headmen, village heads and ward councillors).

Study design and settings

The study analysed data from a cluster randomised trial (CRT) comparing two community-based HIVST distribution models, conducted between 2018 and 2019, in six rural districts (Shamva, Muzarabani, Mutoko, Zvimba, Shurugwi and Umguza) in five provinces of Zimbabwe where Population Service International (PSI) was implementing HIV testing services. Detailed information of the CRT was described elsewhere [22].

Briefly, in the CRT, clusters were defined as headman units, which are administrative units in rural areas of Zimbabwe responsible for implementing community level activities. Forty clusters were randomly allocated to either community-led or paid HIVST distribution arms using 1:1 restricted randomisation. To be considered potentially eligible, clusters had to have at least three census enumeration areas (EAs) within its boundaries, not share a health facility with a neighbouring cluster and the distance between any two selected clusters was at least 20

km. Randomisation was restricted on district, distance from the centre of the headman unit to the nearest health facility and availability of PrEP at the nearest health facility. Consent for communities to participate was sought and obtained from the community leaders.

In the community-led arm, community members at least 16 years of age were invited to the first meeting during which they were introduced to the concept of a community-led model and the importance of HIV testing. The concept of antiretroviral treatment (ART) for preventing HIV transmission within their headman unit and the potential to reduce new HIV infections through early HIV testing and linkage to ART was discussed and explained. Members were entrusted with the responsibility of developing their own HIVST kits distributing models with the options of selecting distributors, deciding on incentives for distributors, managing of kits and how to ensure adherence to regulatory requirements. The distribution models had to adhere to the following MOHCC guidelines testing criteria: kits only to be given to individuals 16 years old and above (age of consent for HIV testing), no forced testing or results disclosure, and confidentiality to be upheld. HIVST distributors received a 3-day training program according to MOHCC guidelines. The training covered areas on HIV testing, facilitating utilization of HIVST kits by others, sharing information to promote and facilitates access to appropriate post-test services, providing information on effectiveness of ART for preventing HIV, and how to complete the data collection tool. Distribution of kits was done over a period of 4-6 weeks.

In the paid distributor model, distributors received the same 3-day training as in the community led model and had 4 weeks to conduct door-to-door distribution in one or two villages. PSI provided training, kits, and a one-off stipend of US\$50 to distributors.

In both HIVST distribution models, distributors were expected to give kits to all willing individuals (one kit per person) in their geographical area who met the testing criteria and share information about linkage to all post-test services that testers could access from local clinics.

Study population and data collection

Trial outcomes were evaluated using household survey 4 months post-distribution to determine uptake of HIVST and linkage to post-test services. All eligible individuals (16 years and above) residing in the selected enumeration areas during HIVST distribution, upon giving written consent to participate in the survey, completed a self-administered Audio Computer-Assisted Survey Instrument questionnaire (ACASI) between 08 October 2019 and 30 December 2019. Tablet computers and headsets were used to enhance participants comfort and ability to respond freely. In this analysis, we excluded participants who provided inconsistent answers regarding never having tested for HIV: those who reported having ever had a positive HIV result, despite reporting never having had an HIV test.

Study variables

We defined the main outcome of never having tested for HIV as having answered "No" to the question "Have you ever been tested for HIV including HIVST?". The question was asked to all participants and the response options were: "Yes"/"No". The secondary outcome of no perceived HIV risk was defined as having selected "Not at risk of contracting HIV infection" to the question "What best describes why you haven't tested for HIV?" and then having selected "Not at risk of contracting HIV infection" to the follow up question: "Of the factors you just mentioned, which was the most important reason you have not tested for HIV?". The selected explanatory variables includes: socio-demographic characteristics (age, sex, education, religion, marital status, employment status, steady partner status, household head status, wealth

quintile), perceived health status, participation in decisions (about their health, household expenditure and visiting relatives), community cohesion(social cohesion, critical consciousness, shared concern), stigma(perceived stigma in community, stigma-any negative attitudes, stigma in health care settings), condomless sex in the past three months and optimism on effectiveness of antiretroviral therapy (ART), referred to as "treatment optimism". (Refer to <u>S1</u> <u>Table</u>, for a comprehensive list of explanatory variables and their levels. The next paragraph gives a summary of how certain variables were derived. Six variables had missing data, and the frequency of missing or "declined to answer" combined was low: perceived health status (1.0%), employment status (1.1%), religion (1.4%), steady partner status (1.7%), treatment optimism (5.0%) and wealth quintile (5.1%). Individuals with missing data were excluded only from the analyses in which the respective missing variable was included when investigating factors associated with never having tested for HIV (ie in the regressions).

Construct explanatory variables. The derived variables were: individual level variables (wealth quintile, participation in decisions, stigma (any negative attitude), and treatment optimism) and cluster level variables(community cohesion sub-scales, and perceived stigma community).

The wealth quintile variable was constructed using data on seventeen binary response questions concerning items owned by surveyed households. As done in the primary study by Sibanda E L et al [22] we ran a polychoric principal component analysis on the dummy variables as recommended and supported by the literature [23-25] and then derived household scores based on the first principal component. The scores were then grouped into quintiles of household index and participants within a household were assigned a quintile level for that household [22].

The variable "participation in decisions" was constructed using three questions asking respectively who makes decisions about: health care, major household purchases and visiting participant's relatives. Participants who made decisions either alone or jointly with partners on all three questions were coded as 2 (three decisions), those who decided alone or jointly with partners in any one or two questions were coded as 1 (one/two decisions), while those who indicated not participating in any of the decisions were coded as 0 (zero decisions).

Factor analysis, a variable reduction technique with one factor solution using the principal component factoring (PCF) method was conducted on community cohesion subscales (social cohesion, shared concern, and critical consciousness), stigma scales ("perceived stigma in community", "stigma: any negative attitude (fear and judgement)", "stigma in health care settings") and "treatment optimism". The factor solution was then rotated using the promax method.

The factor scores for community cohesion subscales and "perceived stigma in community" scale were then summarised at a cluster level using the median, and finally the clusters were divided in tertiles and each cluster labelled as either low, medium or high based on the calculated median. The "stigma: any negative attitude (fear and judgement)" and "treatment optimism" scales scores were divided into tertiles at individual level and each individual was labelled respectively as either having low, medium or high stigma and low, medium or high "treatment optimism".

A total of 27 questions were related to community cohesion (See <u>S2 Table</u>). These were grouped into three subscales: social cohesion (6 items), shared concern (10 items) and critical consciousness (11 items). Response options for the items were: strongly agree, somewhat agree, neither agree nor disagree, somewhat disagree, and strongly disagree. The internal consistencies of the subscales as measured by Cronbach alpha were 88% (social cohesion), 92% (shared concern) and 95% (critical consciousness). The community cohesion questions were adopted from the Lipman Study [26]. The variance explained by the first factors of the sub-construct variables was as follows: 64% for "social cohesion", 58% for "shared concern", and 66% for "critical consciousness".

Ten questions related to stigma (see <u>S2 Table</u>) were adapted from the Hargreaves paper and had previously been validated [<u>27</u>]; they were grouped into 3 subscales: "stigma in health care settings" (2 items), "perceived stigma in community" (5 items) and "stigma: any negative attitude (fear and judgement)" (3 items). Response options on a Likert scale were strongly agree, agree, unsure, disagree, and strongly disagree. Cronbach alpha for the sub-scales were 66% ("stigma in healthcare settings"), 78% ("perceived stigma in the community") and 78% (any negative attitude (fear and judgement)). The variance explained by the first factors of the sub-construct variables was, 70% for "stigma: any negative attitude (fear and judgement)" and 53% for "perceived stigma in the community".

The "treatment optimism" variable was measured through four questions obtained from WHO generic tools for operations research [28] (see <u>S2 Table</u>), rated on the same 5-point Likert scale as the questions related to stigma. The Cronbach alpha was 75%, with the first factor explaining 58% of the variance.

Cronbach alpha and Kaiser-Meyer Olkin (KMO) measure of sampling adequacy (MSA) values for the constructs were all within the acceptable range of 70% to 100% [29–31], except for "stigma in health care settings" where both the Cronbach alpha(66%) and the KMO MSA (50%) fell below the acceptable range. As a result, "stigma in health care settings" was excluded from the analysis.

See <u>S2 Table</u>; for further details on community cohesion, stigma, and "treatment optimism" scales.

Inclusivity in global research. Additional information regarding the ethical, cultural, and scientific considerations specific to inclusivity in global research is included in <u>S1 Text</u>.

Data analysis

Categorical variables were described using frequencies and proportions. The median (IQR) was used to summarise the age of respondents. Unadjusted and adjusted mixed effects logistic regression with a random intercept at headman unit level (to account for clustering within each headman unit) was used to model the relationship between independent variables and the outcome. We first examined each factor separately and then we considered two adjustments: adjustment for sex, and age (Model I) and adjustment for sex, age, religion, and education (Model II). This strategy was done to reduce the risk of over adjustment. We presented the results from the two models to show the effect of additional adjustment for religion and education. The "perceived stigma in community" variable and community cohesion subscale variables were used as cluster level predictors in the analysis, while other variables were used as individual level predictors. We also tested for interaction (using likelihood ratio test) between sex and condomless sex in the past 3 months, to assess whether the effect of condomless sex in the past 3 months on never having tested for HIV varied between men and women.

Using the same independent variables, we also analysed factors associated with no perceived HIV risk among participants who reportedly never tested for HIV. Results are presented as unadjusted and adjusted odds ratios with 95% confidence intervals. All statistical analyses were performed using STATA version 17.

Results

Of the 11,150 survey participants recruited to the community-led CRT, 74 (<1%) were excluded from the analysis as they provided inconsistent replies regarding the outcome of interest (never tested for HIV but reported an HIV positive result).

<u>Table 1</u> presents the characteristics of study participants included in the analysis. The median (IQR) age of participants was 32 (22,45) years and 54.5% were women. Nearly half of

	Ever tested	Never tested	Total	
All	[9254/11076]	[1822/11076]	11076	
	83.5%	16.5%	(100%)	
Population Characteristics	Toos (1 oo	[]		
Age (years),[n] median(IQR)	[9254] 33	[1822] 25	[11076] 32	
Age category	())		(,)	
16-24 years, n (%)	2518 (73.8%)	894 (26.2%)	3412	
10 21 yours, 11 (70)	2010 (701070)	0, 1 (20,2,0)	(30.8%)	
25-34 years, n (%)	2357 (89.4%)	281 (10.6%)	2638 (23.8%)	
35-44 years, n (%)	2014 (91.7%)	183 (8.3%)	2197 (19.8%)	
45 + years, n (%)	2365 (83.6%)	464 (16.4%)	2,829 (25.5%)	
Sex				
Male, n (%)	4003 (79.5%)	1034 (20.5%)	5037 (45.5%)	
Female, n (%)	5251 (87.0%)	788 (13.0%)	6039 (54.5%)	
Household head status	;	'	'	
Household head, n (%)	3840 (86.5%)	601 (13.5%)	4441 (40.1%)	
Household head rep, n (%)	1062 (89.3%)	127 (10.7%)	1189 (10.7%)	
Not household head/rep, n (%)	4352 (79.9%)	1094 (20.1%)	5446 (49.2%)	
Education				
Primary Complete/less, n (%)	3630 (83.5%)	718 (16.5%)	4348 (39.3%)	
Some secondary, n (%)	2209 (79.7%)	552 (20.3%)	2794 (25.2%)	
Secondary complete/tertiary, n (%)	3397 (86.4%)	537 (13.6%)	3934 (35.5%)	
Employment				
Unemployed, n (%)	7082 (82.9%)	1465 (17.1%)	8547 (77.2%)	
Self-employed/subsistence farmer, n (%)	1031 (86.2%)	165 (13.8%)	1196 (10.8%)	
Formally employed, n (%)	1053 (87.3%)	153(12.7%)	1206 (10.9%)	
Missing, n (%)	88(69.3%)	39 (30.7%)	127(1.1%)	
Religion				
Apostolic, n (%)	3636 (85.9%)	599 (14.1%)	4235 (38.2%)	
Catholic/ Protestant, n (%)	2071 (81.9%)	458 (18.1%)	2529 (22.8%)	
Pentecostal, n (%)	1145 (87.1%)	169 (12.9%)	1314 (11.9%)	
No religion/ATR, n (%)	1421 (79.5%)	367 (20.5%)	1788 (16.1%)	
Other, n (%)	981 (81.1%)	229 (18.9%)	1210 (10.9%)	

Table 1. Descriptive characteristics of participants (n=11,076).

(Continued)

Table I. (Commuted)

	Ever tested	Never tested	Total
Marital status			
First marriage (currently), n (%)	3636 (89.5%)	599 (10.5%)	5272 (47.6%)
Remarried after divorce/wid- owed, n (%)	2071 (90.9%)	458 (9.1%)	1449 (13.1%)
Previously married (Widowed/ Separated/Divorced), n (%)	1145 (85.2%)	169 (14.8%)	1626 (14.9%)
Never married, n (%)	1421 (67.3%)	367 (32.7%)	2576 (23.3%)
Missing, n(%)	101 (66.0%)	52 (33.0%)	153 (1.4%)
Current steady partner			
No, n (%)	2,792(72.9%)	1,039 (27.1%)	3831 (34.6%)
Yes, n (%)	6329 (89.7%)	726 (10.3%)	7055 (63.7%)
Missing, n(%)	133 (70.0%)	57 (30.0%)	190 (1.7%)
Health status:			
Very good, n (%)	2606 (81.8%)	578 (18.2%)	3184 (28.8%)
Good, n (%)	2959 (82.4%)	631 (17.6%)	3590 (32.4%)
Fair, n (%)	2493 (85.4%)	425 (14.6%)	2918 (26.4%)
Poor, n (%)	1118 (87.8%)	156 (12.2%)	1274 (11.5%)
Missing, n(%)	72 (70.9%)	32 (29.1%)	110 (1.0%)
Wealth quintile:			
Lowest, n (%)	1572 (83.0%)	321 (17.0%)	1893 (17.1%)
Second, n (%)	1739 (85.9%)	285 (14.1%)	2024 (18.3%)
Middle, n (%)	1736 (83.0%)	354 (16.9%)	2090 (18.9%)
Fourth, n (%)	1815 (84.0%)	345 (16.0%)	2160 (19.5%)
Highest, n (%)	1937 (82.7%)	405 (17.3%)	2342 (21.1%)
Missing, n(%)	455 (80.3%)	112 (19.7%)	567 (5.1%)
Number of decisions participated in	(household purchases, healt	h care and family visits)	
Three, n (%)	5321 (86.8%)	807(13.2%)	6125 (55.3%)
One/two, n (%)	2802(84.9%)	497(15.1%)	3299 (29.8%)
None, n (%)	1131(68.6%)	518 (31.4%)	1649 (14.9%)
Condomless sex in the past 3 months			
Yes, n (%)	5861 (89.9%)	655(10.1%)	6516 (58.8%)
No, n (%)	3393 (74.4%)	1167(25.6%)	4560 (41.2%)
Social cohesion*:			

(Continued)

	Ever tested	Never tested	Total
Low, n (%)	2492(85.8%)	413(14.2%)	2905 (26.2%)
Medium, n (%)	3037(82.1%)	663(17.9%)	3700 (33.4%)
High, n (%)	3725(83.3%)	746(16.7%)	4471 (40.4%)
Critical consciousness*:			
Low, n (%)	2970(83.3%)	596(16.7%)	3566 (32.2%)
Medium, n (%)	3142(83.8%)	609(16.2%)	3751 (33.9%)
High, n (%)	3142 (83.6%)	617(16.4%)	3759 (33.9%)
Shared concern*:			
Low, n (%)	3042 (83.3%)	612 (16.7%)	3654 (33.0%)
Medium, n (%)	3089 (83.2%)	622 (16.8%)	3711 (33.5%)
High, n (%)	3123 (84.2%)	588 (15.8%)	3711 (33.5%)
Perceived stigma in communi	ty*:		
Low, n (%)	1409(84.4%)	261(15.6%)	1670 (15.1%)
Medium, n (%)	4683(83.9%)	898(16.1%)	5581(50.4%)
High, n (%)	3162(82.7%)	663(17.3%)	3825 (34.5%)
Stigma: Any negative attitude			
Low, n (%)	2689(86.2%)	430(13.8%)	3119 (28.2%)
Medium, n (%)	3517(85.4%)	601(14.6%)	4118(37.2%)
High, n (%)	3048(79.4%)	791(20.6%)	3839 (34.7%)
Treatment optimism:			
Low, n (%)	2880 (82.1%)	627 (17.9%)	3507 (31.6%)
Medium, n (%)	2882 (82.2%)	626 (17.8%)	3508 (31.7%)
High, n (%)	3080 (87.8%)	428 (12.2%)	3508 (31.7%)
Missing, n(%)	412 (74.5%)	141 (25.5%)	553 (5.0%)

Table 1. (Continued)

*Cluster level variables.

https://doi.org/10.1371/journal.pgph.0004459.t001

the participants (50.8%) were either household heads or household head representatives, with 40.1% being household heads. Three fifths (60.7%) had at least some secondary education level and 39.3% had at-most primary education level. More than three quarters (78.1%) were unemployed, whilst 11.1% were formally employed and 10.9% were self-employed. In terms of religion, 38.2% were of the apostolic faith, about a quarter (22.8%) were either Catholics or Protestants,16.1% were either of African tradition religion (ATR) or not religious, whilst Pentecostals and other religions constituted just over 10.0% each. In terms of marital status: almost



Factors associated with never testing for HIV in the adjusted analysis



https://doi.org/10.1371/journal.pgph.0004459.g001

half of the participants (48.3%) were currently married in first marriage, a quarter were never married (23.6%), 13.3% had remarried after divorce or death of a partner and the remaining 14.9% were previously married (widowed/divorced). Overall, two thirds (64.8%) had a steady partner and about six in ten reported at least good health status (61.7%). Seventeen percent (1,822/11,076) of participants reported they had never had an HIV test.



Fig 2. Most important reason for not testing for HIV.

https://doi.org/10.1371/journal.pgph.0004459.g002

Fig 1 and S3 Table present findings from the unadjusted and adjusted analysis. In the unadjusted analysis, all variables except "critical consciousness", "shared concern" and "perceived stigma in community" were significantly associated with the odds of never having tested for HIV (p<0.05). In Model I (adjusted for sex and age), being male (adjusted odds ratio [AOR]=1.69; 95% Confidence Interval [CI]=1.52-1.87); younger or older than 35-44 years: 16-24 years (AOR=3.84; 95% CI=3.23-4.55), 25-34 years (AOR=1.30; 95% CI=1.07-1.59), and at least 45 years (AOR=2.17; 95% CI=1.80-2.60), compared to the middle age group 35-44 years; having lower levels of education: at most primary (AOR=1.68; 95% CI=1.46-1.98) and some secondary (AOR=1.62; 95% CI=1.42-1.86), compared to have at least completed secondary; being unemployed (AOR=1.39; 95% CI=1.15-1.69), compared to the formally employed; being never married (AOR=3.48; 95% CI=2.98-4.07) or widowed/separated/ divorced (AOR=1.41; 95% CI=1.19-1.68), compared to being currently in the 1st marriage; with high level of HIV stigma (any negative attitudes) (AOR=1.42; 95% CI=1.24-1.62), compared to low level; with low or medium levels of treatment optimism on effectiveness of ART (respectively AOR=1.52,95%CI=1.32-1.74 and AOR=1.53,95%CI=1.33-1.75), compared to high level; being neither household-head nor household-head representative (OR=1.38,95%CI=1.21-1.57), compared to household heads; having a lower participation in any of the three areas of household decisions (AOR=2.07; 95% CI=1.79-2.39), compared to participation in all three areas; reporting no condomless sex in the past 3 months (AOR=2.58; 95% CI=231-2.87) were more likely to be never testers.

When testing the interaction between sex and condomless sex in the past 3 months, we found that the association between reporting no condomless sex and never testing was greater for women compared to men: AOR=3.53 (95% CI=2.99-4.16) vs AOR=1.99 (95% CI=1.71-2.30; p value for interaction<0.001). When additionally adjusting all associations for education

Table 2. Factors associated with no perception of HIV risk among participants who reportedly never tested for HIV.

	Unad- justed OR	95% CI	Overall P value	Adjusted ORª	95% CI	Overall P value
Population Characteristics						
Age category			0.024			< 0.001
16-24 years	1.60	[1.15, 2.23]		1.59	[1.14, 2.22]	
25-34 years	1.00	[0.68, 1.47]		0.99	[0.67, 1.46]	
35-44 years	1.0	Ref		1.0	Ref	
45 + years	2.30	[1.61, 3.28]		2.26	[1.58, 3.24]	
Sex			0.131			0.495
Female	1.0	Ref		1.0	Ref	
Male	0.86	[0.72, 1.04]		0.93	[0.77, 1.13]	
Household head status:			0.519			0.498
Household head	1.0	Ref		1.0	Ref	
Household head rep	1.13	[0.76, 1.66]		1.19	[0.79, 1.79]	
Not household head/rep	1.10	[0.90, 1.34]		1.15	[0.90, 1.46]	
Education level:			< 0.001			0.006
Secondary complete/tertiary	1.0	Ref		1.0	Ref	
Some secondary	1.13	[0.88, 1.44]		1.06	[0.83, 1.35]	
Primary complete/less	1.63	[1.30, 2.06]		1.46	[1.14, 1.87]	
Employment status:			0.015			0.193
Formally employed	1.0	Ref		1.0	Ref	
Self employed/subsistence farmer	1.32	[0.84, 2.07]		1.23	[0.77, 1.94]	
Not employed	1.52	[1.07, 2.15]		1.37	[0.96, 1.97]	
Religion:			0.033			0.255
Apostolic	1.0	Ref		1.0	Ref	
Catholic/ Protestant	1.15	[0.90, 1.48]		1.13	[0.88 1.46]	
Pentecostal	1.06	[0.75, 1.50]		1.10	[0.77, 1.57]	
No religion/ ATR	1.15	[0.88, 1.50]		1.19	[0.89, 1.56]	_
Moslem & Other	1.49	[1.09, 2.04]		1.44	[1.05, 1.97]	_
Marital Status:			0.112			0.237
1st marriage/staving as married	1.0	Ref		1.0	Ref	
Remarried after divorce/widowed	1.00	[0.68,1.47]		1.11	[0.74, 1.64]	_
Widowed/Separated/Divorced	1.28	[0.94, 1.74]		0.99	[0.72,1.38]	
Never married	1.18	[0.95, 1.46]		1.37	[1.01, 1.85]	_
Current steady partner			< 0.001			0.003
Yes	1.0	Ref		1.0	Ref	
No	1.42	[1.17, 1.72]		1.37	[1.12, 1.69]	_
Perceived health status:		[,]	<0.001		[,,]	< 0.001
Very good	1.47	[1.02, 2.10]		1.78	[1.21, 2.63]	
Good	1.07	[0.75, 1.53]		1.24	[0.85, 1.79]	_
Fair	0.95	[0.66.1.38]		1.03	[0.70, 1.51]	
Poor	1.0	Ref		1.0	Ref	
Household asset quintile:	1.0		0.115	1.0		0.318
Highest	1.0	Ref	0.115	1.0	Ref	0.010
Lowest	0.79	[0.59 1.08]		0.75	[0.55, 1.02]	
Second	0.86	[0.63 1 18]		0.85	[0.62 1 17]	
Middle	0.78	[0.58 1.05]		0.03	[0.57 1.03]	
Fourth	0.02	[0.69 1.25]		0.89	[0.66, 1.20]	
i vui tii	0.95	[0.09, 1.23]		0.07	[0.00, 1.20]	

(Continued)

Table 2. (Continued)

Unad- justed OR	95% CI	Overall P value	Adjusted ORª	95% CI	Overall P value
		0.439			0.751
1.0	Ref		1.0	Ref	
0.95	[0.75, 1.19]		0.98	[0.76, 1.25]	
1.11	[0.88, 1.39]		1.07	[0.83, 1.40]	
		< 0.001			0.002
1.0	Ref		1.0	Ref	
1.44	[1.18, 1.75]		1.40	[1.13, 1.72]	
		0.724			0.331
1.0	Ref		1.0	Ref	
1.27	[0.92, 1.74]		1.25	[0.89, 1.73]	
1.09	[0.80, 1.50]		1.03	[0.75, 1.42]	
		0.340			0.357
1.17	[0.85, 1.60]		1.27	[0.92, 1.75]	
1.02	[0.74, 1.38]		1.14	[0.83, 1.56]	
1.0	Ref		1.0	Ref	
		0.159			0.159
1.25	[0.91, 1.70]		1.36	[0.99, 1.86]	
1.07	[0.78, 1.46]		1.11	[0.81, 1.52]	
1.0	Ref		1.0	Ref	
		0.014			0.051
1.0	Ref		1.0	Ref	
1.14	[0.80, 1.62]		1.13	[0.78, 1.62]	
1.51	[1.04, 2.20]		1.51	[1.02, 2.22]	
		0.231			0.069
1.0	Ref		1.0	Ref	
0.87	[0.68, 1.12]		0.89	[0.69, 1.15]	
1.11	[0.87, 1.41]		1.15	[0.90, 1.47]	
		0.458			0.198
0.89	[0.69, 1.14]		0.87	[0.61, 1.10]	
0.79	[0.61, 1.01]		0.79	[0.64, 1.07]	
1.0	Ref		1.0	Ref	
	Unad- justed OR 1.0 0.95 1.11 1.0 1.44 1.0 1.27 1.09 1.17 1.02 1.0 1.27 1.09 0.117 1.02 1.0 1.17 1.02 1.01 1.02 1.00 0.1.14 1.51 0.87 1.11 0.89 0.79 1.0	Unad- justed OR 95% CI 1.0 Ref 0.95 [0.75, 1.19] 1.11 [0.88, 1.39] 1.11 [0.88, 1.39] 1.0 Ref 1.0 Ref 1.11 [0.88, 1.39] 1.11 [0.88, 1.39] 1.0 Ref 1.44 [1.18, 1.75] 1.0 Ref 1.27 [0.92, 1.74] 1.09 [0.80, 1.50] 1.17 [0.85, 1.60] 1.02 [0.74, 1.38] 1.0 Ref 1.02 [0.74, 1.38] 1.0 Ref 1.02 [0.74, 1.38] 1.0 Ref 1.14 [0.80, 1.62] 1.51 [1.04, 2.20] 1.11 [0.87, 1.41] 0.87 [0.68,	Unad- justed OR 95% CI Overall P value 0.439 0.439 1.0 Ref 0.439 0.95 [0.75, 1.19] 1 1.11 [0.88, 1.39] - 1.11 [0.88, 1.39] - 1.11 [0.88, 1.39] - 1.11 [0.88, 1.39] - 1.0 Ref - 1.44 [1.18, 1.75] 0.724 1.0 Ref - 1.27 [0.92, 1.74] - 1.09 [0.80, 1.50] - 1.09 [0.85, 1.60] - 1.02 [0.74, 1.38] - 1.02 [0.74, 1.38] - 1.02 [0.74, 1.38] - 1.01 Ref - 1.02 [0.74, 1.38] - 1.01 Ref - 1.02 [0.78, 1.46] - 1.01 Ref - 1.01 Ref - 1.10	Unad- justed OR 95% CI Overall P value Adjusted OR* 1.0 Ref 0.439 1.0 1.0 Ref 1.0 0.98 1.11 [0.88, 1.39] 0.98 1.07 1.11 [0.88, 1.39] 1.07 0.98 1.11 [0.88, 1.39] 1.07 0.98 1.11 [0.88, 1.39] 1.07 0.98 1.0 Ref 1.0 1.07 1.0 Ref 1.0 1.40 1.27 [0.92, 1.74] 1.25 1.09 1.27 [0.92, 1.74] 1.25 1.03 1.17 [0.85, 1.60] 1.03 1.14 1.09 [0.80, 1.50] 1.13 1.14 1.02 [0.74, 1.38] 1.14 1.0 1.02 [0.74, 1.38] 1.14 1.0 1.01 Ref 1.0 1.11 1.0 Ref 1.0 1.11 1.0 Ref 1.0 1.11 1.0 <td>Unad- justed OR 95% CI Overall Pvalue Adjusted OR 95% CI 1.0 Ref 0.439 1.0 Ref 1.0 Ref 0.439 0.98 [0.75, 1.2] 0.95 [0.75, 1.19] 0.98 [0.76, 1.25] 1.11 [0.88, 1.39] 1.07 [0.83, 1.40] 1.0 Ref 1.0 [0.83, 1.40] 1.0 Ref 1.0 [0.83, 1.40] 1.44 [1.18, 1.75] 1.40 [1.13, 1.72] 1.44 [1.18, 1.75] 1.40 [0.89, 1.73] 1.44 [1.18, 1.75] 1.40 Ref 1.47 [0.92, 1.74] 1.25 [0.89, 1.73] 1.00 Ref 1.0 Ref 1.17 [0.85, 1.60] 1.27 [0.92, 1.75] 1.02 [0.74, 1.38] 1.14 [0.83, 1.56] 1.01 Ref 1.0 Ref 1.02 [0.91, 1.70] 1.14 [0.81, 1.5] 1.02 [0.91, 1.70] 1.11 <t< td=""></t<></td>	Unad- justed OR 95% CI Overall Pvalue Adjusted OR 95% CI 1.0 Ref 0.439 1.0 Ref 1.0 Ref 0.439 0.98 [0.75, 1.2] 0.95 [0.75, 1.19] 0.98 [0.76, 1.25] 1.11 [0.88, 1.39] 1.07 [0.83, 1.40] 1.0 Ref 1.0 [0.83, 1.40] 1.0 Ref 1.0 [0.83, 1.40] 1.44 [1.18, 1.75] 1.40 [1.13, 1.72] 1.44 [1.18, 1.75] 1.40 [0.89, 1.73] 1.44 [1.18, 1.75] 1.40 Ref 1.47 [0.92, 1.74] 1.25 [0.89, 1.73] 1.00 Ref 1.0 Ref 1.17 [0.85, 1.60] 1.27 [0.92, 1.75] 1.02 [0.74, 1.38] 1.14 [0.83, 1.56] 1.01 Ref 1.0 Ref 1.02 [0.91, 1.70] 1.14 [0.81, 1.5] 1.02 [0.91, 1.70] 1.11 <t< td=""></t<>

a-Adjusted for age and sex, CI- confidence interval; OR-Odds ratio.

https://doi.org/10.1371/journal.pgph.0004459.t002

and religion (Model II), the results did not change in any meaningful way, see $\underline{Fig 1}$ and/or $\underline{S3}$ Table.

Among participants who reported never having tested for HIV (n=1,822), the most important cited reason for never testing was no perceived HIV risk (50%) (of which a third reportedly engaged in condomless sex), followed by being afraid of testing HIV positive and not feeling sick enough to test for HIV, both 8% each as shown in Fig.2.

Factors associated with increased odds of reporting no HIV risk perception in the adjusted analysis (adjusted for sex and age) were low education levels; never married, compared to being currently in the 1st marriage; younger or older age: 16-24 years, and at-least 45 years; not having a steady partner and not reporting condomless sex, compared to those who reportedly had. Also, reporting very good health status was associated with increased odds of reporting no HIV risk perception compared with poor health, see <u>Table 2</u>.

Discussion

We found that 17% of the participants had never had an HIV test, in rural communities in Zimbabwe between 2018-2019 despite a community-based distribution of HIV self-testing kits, which greatly improved access to HIV testing [32,33]. This study did not collect baseline information on HIV testing uptake in the communities and did not evaluate whether the introduction of community-based HIV self-testing increases testing rates. However, previous evidence had reported that HIVST increases uptake of HIV testing [32,33]. It is also important to provide a range of HIVST delivery models, as testing rates have been shown to improve with a variety of HIVST models and support tools [34]. These results are comparable with the findings from a household survey done in Zimbabwe in 2020 [7] which reported that 20% and 16% in rural and urban areas respectively had never tested for HIV. More needs to be done to identify non HIV testing sub-populations if the country is to meet the UNAIDS 2030 target of ending AIDS as a public health threat [5].

Never testers were more likely to be men, and this is in line with several other studies and reports [35–39]. Possible reasons are that men tend to have poorer health seeking behaviour [40,41], use partner's HIV status as substitute for testing, and that women access HIV information and testing through antenatal clinics [42]. High uptake among women has also been attributed to the integration of HIV testing into clinical services for women [39]. In this study, HIV self-testing, which was found to be highly acceptable among men [43], was widely available so our findings need to be interpreted in that context.

Our study supports findings from several studies which reported an association between age and never having tested for HIV, with younger people having higher odds of being never testers for HIV [35,36,44], however, they may be less likely to be sexually active. Young people face barriers to HIV testing and these include low HIV risk perception [45], scarce opportunities for accessing HIV testing services, not trusting in HIV testing services, and unfriendly attitudes by health care workers [46]. Among never testers, the most cited reason was no perceived HIV risk: young people (15-24 years) and older people (45+ years) were more likely to report no HIV risk perception when compared to the 35-44 years old. Considering that HIV self-test kits were distributed in the communities four weeks prior to the survey, there were plenty opportunities for accessing HIV services. Other possible explanations are fear of stigma and discrimination related to HIV testing and fear of a positive HIV result [47]. We also found that individuals aged 45 years and above, had higher odds of being never testers compared to 35-44 years age group. This is likely attributed to low HIV risk perception among the older age group as reported in this study. Studies among young people have reported a positive correlation between HIV testing uptake and risk perception [48,49]. However, the reason why those at least 45 years were unlikely to test for HIV needs to be investigated further.

In this study, lower levels of education were associated with never having tested for HIV, and this finding also emerged in studies conducted in Ethiopia [50] on HIV testing practice among women 15-24 years, in Zimbabwe [51] and several other studies [35,36,44]. Greater knowledge of HIV allows for better understanding of the benefits of testing for HIV [52] and therefore greater uptake of HIV testing [53].

Consistent with other studies [54,55], the unemployed were more likely to be never testers compared to the formally employed. Research suggests that employment is associated with increased social connections, financial freedom, and independence [56]. However, as part of this trial HIV self-testing kits were distributed for free, with instructions and demonstrations on how to use the kits. Post-test services information was shared, and recipients would test at their own convenient time without disclosing results to distributors. Therefore, further investigations are needed to understand the reasons why the unemployed were less likely to test.

Never married participants were more likely to be never testers than those who were married, even after adjusting for age. This result supports finding from previous studies in Zimbabwe, Kenya, and Ethiopia [36,44,51]. Similarly, those who were currently not in a partnership but who were previously (widowed/ separated/ divorced) were more likely to be never testers than married participants. A possible explanation is that participants who are currently not in a partnership, like those never married, are at low risk of contracting HIV [48,49].

Stigma (any negative attitudes) was associated with being a never tester. In line with our findings, previous studies have reported lower HIV testing uptake among participants with high levels of stigma [36,57,58]. Interventions designed to reduce individual stigma may help in improving uptake of HIV testing. Owing to the complexity of HIV/AIDS related stigma, efforts to reduce this stigma must address the different aspects of it which varies by culture and on whether its intrapersonal or societal [59]. Offering HIV testing as a package together with other health services such as routine medical check-ups can contribute to reduce stigma associated with HIV [60]. In addition, other ways to reduce HIV related stigma and discrimination include, but is not limited to, strengthening capacity of community health workers by ensuring appropriate linkages between communities and formal health systems; ensuring that young people have access to youth-friendly HIV services and comprehensive sexuality education; engaging communities in stigma- and discrimination-reduction activities; educating workplace communities on HIV, comorbidities and legal literacy to promote positive social norms related to HIV, and routinely assessing knowledge, attitudes and practices of health-care workers towards vulnerable populations to support health facility administrators to identify and address any issues [61].

Lower levels of treatment optimism were associated with never having tested for HIV. One study reported that increased "treatment optimism" was associated with having condomless sex [62] and in this study we found an association between condomless sex in the past 3 months and HIV testing, with the likelihood of testing being higher among those who reported condom less sex. There is need to investigate further how "treatment optimism" affects HIV testing.

Study limitations

We acknowledge our study has some limitations. Firstly, since information on HIV testing, stigma, health status among other variables is self-reported there is a possibility of social desirability bias. We endeavoured to minimise this by using ACASI. Secondly, data was collected from only 6 rural districts, and this might limit generalizability of findings to other districts especially non rural populations. Also, due to the cross-sectional nature of the study, causality between the outcome and some independent variables is difficult to assess. Another limitation is the exclusion of (74/11,150) 0.66% participants with inconsistency responses. However, the proportion is relatively small such that it is unlikely to have an impact on the validity of the results. Lastly, data analysed was collected in 2018-19, thus the prevalence of never testing has likely decreased in the meantime. This study focusses on the general population, however it would be important to understand whether this is similar among key populations which are an important group contributing to the HIV testing gap.

Conclusion

We found that never testers were more likely to be males, of age below 35 or above 45, with low level of education, unemployed, never married, those projecting high level of stigma, lower level of HIV treatment optimism, those who were not participating in decision making. As the country aims to meet the 2030 UNAIDS global targets, MOHCC need to scale up acceptable and targeted interventions to improve HIV testing in different subpopulations. The interventions include scaling up distribution of information, education and communication materials on HIV related topics in schools and communities, in local languages and through social media and community campaigns to reduce stigma associated with testing for HIV and targeting men and young people through initiatives such as men's clinics and workplace programs. Provision of knowledge on where HIV testing services can be found through media and mobile HIV testing campaigns has been reported to increase uptake of testing among men and young people especially in rural areas [63]. It is also crucial to increase HIVST delivery models both at facilities and communities, as this enhances testing uptake by offering more choices to clients [34].

Supporting information

S1 Table. Explanatory variables and their levels. (DOCX)

S1 Text. Inclusivity in Global Research Checklist. (DOCX)

S2 Table. Derivation of construct variables. (DOCX)

S3 Table. Risk factors associated with never testing for HIV. (DOCX)

S4 Table. Descriptive characteristics of those reporting no perception of HIV risk. (DOCX)

S1 Data. Data set. (DTA)

S1 File. Data dictionary. (PDF)

Acknowledgments

We acknowledge the study participants, community leaders, study staff and Ministry of Health and Child Care.

Author contributions

- **Conceptualization:** Wellington Murenjekwa, Frances M Cowan, Cheryl Johnson, Amon Mpofu, Getrude Ncube, Owen Mugurungi, Karin Hatzold, Elizabeth L Corbett, Euphemia Sibanda, Valentina Cambiano.
- Formal analysis: Wellington Murenjekwa, Fiona C Lampe, Andrew N Phillips, Valentina Cambiano.
- **Funding acquisition:** Frances M Cowan, Karin Hatzold, Elizabeth L Corbett, Valentina Cambiano.
- **Methodology:** Wellington Murenjekwa, Frances M Cowan, Karin Hatzold, Elizabeth L Corbett, Euphemia Sibanda.

Project administration: Kudzai Chidhanguro, Euphemia Sibanda, Valentina Cambiano.

Software: Wellington Murenjekwa.

Supervision: Frances M Cowan, Fiona C Lampe, Andrew N Phillips, Euphemia Sibanda, Valentina Cambiano.

Visualization: Wellington Murenjekwa.

- Writing original draft: Wellington Murenjekwa.
- Writing review & editing: Kudzai Chidhanguro, Frances M Cowan, Fiona C Lampe, Cheryl Johnson, Amon Mpofu, Getrude Ncube, Owen Mugurungi, Karin Hatzold, Elizabeth L Corbett, Andrew N Phillips, Euphemia Sibanda, Valentina Cambiano.

References

- 1. WHO. WHO Key Facts HIV 2021 World Health Organisation; 2022.
- Ayieko J, Charlebois ED, Maeri I, Owino L, Thorp M, Bukusi EA, et al. Improving care engagement for mobile people living with HIV in rural western Kenya. PLoS One. 2023;18(11):e0288087. <u>https://doi.org/10.1371/journal.pone.0288087</u> PMID: <u>37992063</u>
- UNAIDS. IN DANGER: UNAIDS Global AIDS Update 2022. Geneva: Joint United Nations Programme on HIV/AIDS; 2022. Licence: CC BY-NC-SA 3.0 IGO.; 2022.
- Population Reference Bureau. 2022 World Population data Sheet. https://2022-wpdsprborg/ [Internet]. 2022 [cited 2023. Available from: <u>https://www.prb.org/resources/2022-world-population-data-sheet-booklet/</u>
- United Nations. Transforming our World: The 2030 Agenda for Sustainable Development. 2015. Contract No.: A/RES/70/1.
- Zimbabwe National Statistics Agency, International ICF. Zimbabwe demographic and health survey, 2010-11. Harare, Zimbabwe Calverton, Md., USA: Zimbabwe National Statistics Agency ICF International; 2012.
- 7. Ministry of Health and Child Care (MoHCC). Zimbabwe Population-based HIV Impact Assessment 2020 (ZIMPHIA 2020). Harare: MoHCC; December 2021.; 2020.
- Gregson S, Mugurungi O, Eaton J, Takaruza A, Rhead R, Maswera R, et al. Documenting and explaining the HIV decline in east Zimbabwe: the Manicaland General Population Cohort. BMJ Open. 2017;7(10):e015898. <u>https://doi.org/10.1136/bmjopen-2017-015898</u> PMID: <u>28988165</u>
- 9. UNFPA. FACT SHEET HIV Decline in Zimbabwe: Positive Behaviour Change Makes a Difference.
- 10. UNDP. FACT SHEET: TOWARDS ENDING AIDS AS A PUBLIC HEALTH THREAT BY 2030. United Nations Development Programme; 2022 December 2022.
- Ministry of Health and Child Care (MoHCC), National AIDS Council (NAC). EXTENDED ZIMBABWE NATIONAL HIV AND AIDS STRATEGIC PLAN (ZNASP3) 2015 – 2020 In: Ministry of Health and Child Care (MoHCC), editor: Ministry of Health and Child Care (MoHCC) and National AIDS Council (NAC).
- 12. United Nations Development Programme (UNDP). Snapshot, UNDP and Global Fund in Zimbabwe. 2024.
- 13. Joint United Nations Programme on HIV/AIDS (UNAIDS). The path that ends AIDS: UNAIDS Global AIDS Update 2023. Geneva; 2023.
- Musuka G, Dzinamarira T. Targeting those left behind in Zimbabwe's HIV response: A call for decriminalisation of key populations to rapidly achieve 95-95-95 targets. S Afr Med J. 2021;111(5):13287. https://doi.org/10.7196/SAMJ.2021.v111i5.15567 PMID: 34852874
- Chowdhury MT, Bershteyn A, Milali M, Citron D, Nyimbili S, Musuka G, et al. Progress Towards UNAIDS's 95-95-95 Targets in Zimbabwe: Sociodemographic Constraints and Geospatial Heterogeneity. medRxiv. 2023.
- Dzinamarira T, Moyo E, Moyo B, Murewanhema G, Cuadros D, Kouamou V, et al. Strengthening and enhancing national antiretroviral drug resistance surveillance in Zimbabwe-A country that has reached UNAIDS 95-95-95 amongst adults. Front Public Health. 2024;12:1346027. <u>https://doi.org/10.3389/fpubh.2024.1346027</u> PMID: 38420022
- **17.** Ministry of Health and Child Care. Zimbabwe National Guidelines on HIV Testing and Counselling. 2014.
- NMTPAC and AIDS and TB Directorate-MOHCC. Guidelines for Antiretroviral Therapy for the Prevention and Treatment of HIV in Zimbabwe. In: The National Medicine and Therapeutics Policy Advisory Committee (NMTPAC) and The AIDS and TB Directorate MoHaCC, editor. 2016.

- Johnson C, Neuman M, MacPherson P, Choko A, Quinn C, Wong VJ, et al. Use and awareness of and willingness to self-test for HIV: an analysis of cross-sectional population-based surveys in Malawi and Zimbabwe. BMC Public Health. 2020;20(1):779. <u>https://doi.org/10.1186/s12889-020-08855-7</u> PMID: 32450840
- Mugauri DH, Ncube G, Chikondowa I, Dupwa B, Gwashure S, Nyazema L, et al. A Formative Evaluation of Zimbabwe's HIV Testing Services Strategy, 2019-A Cross-Sectional Study. J AIDS Immune Syst. 2020;1(1).
- 21. Central Statistical Office (CSO) and Macro International Inc. Zimbabwe Demographic and Health Survey 2005-06. Calverton, Maryland. 2007.
- 22. Sibanda EL, Mangenah C, Neuman M, Tumushime M, Watadzaushe C, Mutseta MN, et al. Comparison of community-led distribution of HIV self-tests kits with distribution by paid distributors: a cluster randomised trial in rural Zimbabwean communities. BMJ Glob Health. 2021;6(Suppl 4):e005000. https://doi.org/10.1136/bmjgh-2021-005000 PMID: 34275872
- 23. Kolenikov S AG. The Use of Discrete Data in PCA: Theory, Simulations, and Applications to Socioeconomic Indices 2004 01/12/2023 [cited 2023. Available from: <u>https://www.measureevaluation.org/</u> resources/publications/wp-04-85.html
- Kolenikov S, Angeles G. Socioeconomic status measurement with discrete proxy variables: Is principal component analysis a reliable answer? Review of Income and Wealth. 2009;55(1):128–65.
- Martel P, Mbofana F, Cousens S. The polychoric dual-component wealth index as an alternative to the DHS index: Addressing the urban bias. J Glob Health. 2021;11:04003. <u>https://doi.org/10.7189/jogh.11.04003 PMID: 33643634</u>
- Lippman SA, Neilands TB, Leslie HH, Maman S, MacPhail C, Twine R, et al. Development, validation, and performance of a scale to measure community mobilization. Soc Sci Med. 2016;157:127–37. https://doi.org/10.1016/j.socscimed.2016.04.002 PMID: 27085071
- 27. Hargreaves JR, Pliakas T, Hoddinott G, Mainga T, Mubekapi-Musadaidzwa C, Donnell D, et al. HIV Stigma and Viral Suppression Among People Living With HIV in the Context of Universal Test and Treat: Analysis of Data From the HPTN 071 (PopART) Trial in Zambia and South Africa. J Acquir Immune Defic Syndr. 2020;85(5):561–70. <u>https://doi.org/10.1097/QAI.00000000002504</u> PMID: 32991336
- 28. Obermeyer CM BS, Carrieri P, Parsons M, Pulerwitz J, Ruternberg N, Sarna A. HIV testing, treatment and prevention:generic tools for operational research 2009 12/12/2023. Available from: <u>https://knowl-edgecommons.popcouncil.org/cgi/viewcontent.cgi?article=1607&context=departments_sbsr-hiv</u>
- Tavakol M, Dennick R. Making sense of Cronbach's alpha. Int J Med Educ. 2011;2:53–5. <u>https://doi.org/10.5116/ijme.4dfb.8dfd PMID: 28029643</u>
- Nasaireh A. Developing and validating instruments for measurement of organizational culture dimensions for organizational development achievement. International Journal of Multidisciplinary and Current Educational Research (IJMCER). 2020;2(5):168–74.
- **31.** Hoque MMSA, Siddiqui AB, Awang BZ, Tams B. Exploratory factor analysis of entrepreneurial orientation in the context of bangladeshi small and medium enterprises (SMES). European Journal of Management and Marketing Studies. 2018;3(2).
- Johnson CC, Kennedy C, Fonner V, Siegfried N, Figueroa C, Dalal S, et al. Examining the effects of HIV self-testing compared to standard HIV testing services: a systematic review and meta-analysis. J Int AIDS Soc. 2017;20(1):21594. <u>https://doi.org/10.7448/IAS.20.1.21594</u> PMID: <u>28530049</u>
- 33. Hatzold K, Gudukeya S, Mutseta MN, Chilongosi R, Nalubamba M, Nkhoma C, et al. HIV self-testing: breaking the barriers to uptake of testing among men and adolescents in sub-Saharan Africa, experiences from STAR demonstration projects in Malawi, Zambia and Zimbabwe. J Int AIDS Soc. 2019;22 Suppl 1(Suppl Suppl 1):e25244.
- Jamil MS, Eshun-Wilson I, Witzel TC, Siegfried N, Figueroa C, Chitembo L, et al. Examining the effects of HIV self-testing compared to standard HIV testing services in the general population: A systematic review and meta-analysis. EClinicalMedicine. 2021;38:100991. <u>https://doi.org/10.1016/j. eclinm.2021.100991</u> PMID: <u>34278282</u>
- 35. Ng'ambi W, Chiumia KI, Chagoma N, J M-B. Factors Associated with Uptake of HIV Testing in Malawi: A Trend Analysis of the Malawi Demographic and Health Survey Data from 2004 to 2016. J HIV AIDS Res. 2020;2(1).
- Ziraba AK, Madise NJ, Kimani JK, Oti S, Mgomella G, Matilu M, et al. Determinants for HIV testing and counselling in Nairobi urban informal settlements. BMC Public Health. 2011;11:663. <u>https://doi.org/10.1186/1471-2458-11-663</u> PMID: 21861898

- **37.** Giguere K, Eaton J, Marsh K, Johnson L, Johnson C, Ehui E. Trends in knowledge of HIV status and efficiency of HIV testing services in sub-Saharan Africa, 2000-20: a modelling study using survey and HIV testing programme data. Lancet HIV. 2021;8(5):e284–93.
- UNAIDS. Women living with HIV are more likely to access HIV testing and treatment 2020: Joint United Nations Programme on HIV/AIDS (UNAIDS); 2020 [21/11/2024]. Available from: <u>https://www.unaids.org/en/resources/presscentre/featurestories/2020/</u> october/20201005_women-hiv-testing-treatment
- World Health Organisation. IMPROVING MEN'S UPTAKE OF HIV TESTING AND LINKAGE TO SER-VICES: policy brief 2021 [cited 2023. Available from: https://apps.who.int/iris/handle/10665/339620
- 40. Chikovore J, Hart G, Kumwenda M, Chipungu GA, Desmond N, Corbett L. Control, struggle, and emergent masculinities: a qualitative study of men's care-seeking determinants for chronic cough and tuberculosis symptoms in Blantyre, Malawi. BMC Public Health. 2014;14:1053. <u>https://doi.org/10.1186/1471-2458-14-1053</u> PMID: 25301572
- Camlin CS, Ssemmondo E, Chamie G, El Ayadi AM, Kwarisiima D, Sang N, et al. Men "missing" from population-based HIV testing: insights from qualitative research. AIDS Care. 2016;28 Suppl 3(Suppl 3):67–73. <u>https://doi.org/10.1080/09540121.2016.1164806</u> PMID: 27421053
- 42. DiCarlo AL, Mantell JE, Remien RH, Zerbe A, Morris D, Pitt B, et al. "Men usually say that HIV testing is for women": gender dynamics and perceptions of HIV testing in Lesotho. Cult Health Sex. 2014;16(8):867–82. <u>https://doi.org/10.1080/13691058.2014.913812</u> PMID: <u>24854495</u>
- 43. Hamilton A, Thompson N, Choko AT, Hlongwa M, Jolly P, Korte JE, et al. HIV Self-Testing Uptake and Intervention Strategies Among Men in Sub-Saharan Africa: A Systematic Review. Front Public Health. 2021;9:594298. https://doi.org/10.3389/fpubh.2021.594298 PMID: 33681120
- 44. Godif M. Factors Associated with HIV Counseling and Testing among Males and Females in Ethiopia: Evidence from Ethiopian Demographic and Health Survey Data. J AIDS Clin Res. 2015;06(03). <u>https://doi.org/10.4172/2155-6113.1000429</u>
- 45. Sisay S, Erku W, Medhin G, Woldeyohannes D. Perception of high school students on risk for acquiring HIV and utilization of voluntary counseling and testing (VCT) service for HIV in Debreberhan Town, Ethiopia: a quantitative cross-sectional study. BMC Res Notes. 2014;7:518. <u>https://doi.org/10.1186/1756-0500-7-518 PMID: 25112147</u>
- 46. Sonko I, Chung M-H, Hou W-H, Chen W-T, Chang P-C. Predictors of HIV testing among youth aged 15-24 years in The Gambia. PLoS One. 2022;17(2):e0263720. <u>https://doi.org/10.1371/journal.pone.0263720</u> PMID: <u>35180256</u>
- Strauss M, Rhodes B, George G. A qualitative analysis of the barriers and facilitators of HIV counselling and testing perceived by adolescents in South Africa. BMC Health Serv Res. 2015;15:250. <u>https:// doi.org/10.1186/s12913-015-0922-0</u> PMID: 26123133
- Dzinamarira T, Moyo E. Adolescents and young people in sub-Saharan Africa: overcoming challenges and seizing opportunities to achieve HIV epidemic control. Front Public Health. 2024;12:1321068. https://doi.org/10.3389/fpubh.2024.1321068 PMID: 38566795
- 49. Ajayi AI, Mudefi E, Adeniyi OV, Goon DT. Achieving the first of the Joint United Nations Programme on HIV/AIDS (UNAIDS) 90-90-90 targets: understanding the influence of HIV risk perceptions, knowing one's partner's status and discussion of HIV/sexually transmitted infections with a sexual partner on uptake of HIV testing. Int Health. 2019;11(6):425–31. <u>https://doi.org/10.1093/inthealth/ihz056</u> PMID: 31365082
- Diress G, Ahmed M, Adane S, Linger M, Alemnew B. Barriers and Facilitators for HIV Testing Practice Among Ethiopian Women Aged 15-24 years: Analysis of the 2016 Ethiopian Demographic and Health Survey. HIV AIDS (Auckl). 2021;12:963–70. https://doi.org/10.2147/HIV.S280590 PMID: 33447085
- Takarinda KC, Madyira LK, Mhangara M, Makaza V, Maphosa-Mutsaka M, Rusakaniko S, et al. Factors Associated with Ever Being HIV-Tested in Zimbabwe: An Extended Analysis of the Zimbabwe Demographic and Health Survey (2010-2011). PLoS One. 2016;11(1):e0147828. <u>https://doi.org/10.1371/journal.pone.0147828</u> PMID: 26808547
- Bhattarai N, Bam K, Acharya K, Thapa R, Shrestha B. Factors associated with HIV testing and counselling services among women and men in Nepal: a cross-sectional study using data from a nationally representative survey. BMJ Open. 2021;11(12):e049415. <u>https://doi.org/10.1136/ bmjopen-2021-049415</u> PMID: <u>34862281</u>
- **53.** Awopegba OE, Ologunowa TO, Ajayi AI. HIV testing and self-testing coverage among men and women in South Africa: an exploration of related factors. 2021(1365-3156 (Electronic)).
- 54. Venkatesh KK, Madiba P, De Bruyn G, Lurie MN, Coates TJ, Gray GE. Who gets tested for HIV in a South African urban township? Implications for test and treat and gender-based prevention

interventions. J Acquir Immune Defic Syndr. 2011;56(2):151–65. <u>https://doi.org/10.1097/QAI.</u> 0b013e318202c82c PMID: 21084993

- Peltzer K, Matseke G. Determinants of HIV testing among young people aged 18 24 years in South Africa. Afr Health Sci. 2013;13(4):1012–20. https://doi.org/10.4314/ahs.v13i4.22 PMID: 24940326
- 56. Maulsby CH, Ratnayake A, Hesson D, Mugavero MJ, Latkin CA. A Scoping Review of Employment and HIV. AIDS Behav. 2020;24(10):2942–55. <u>https://doi.org/10.1007/s10461-020-02845-x</u> PMID: <u>32246357</u>
- **57.** Adugna DG, Worku MG. HIV testing and associated factors among men (15-64 years) in Eastern Africa: a multilevel analysis using the recent demographic and health survey. BMC Public Health. 2022;22(1):2170. https://doi.org/10.1186/s12889-022-14588-6 PMID: 36434555
- Mahlalela NB, Manne-Goehler J, Ohene-Kwofie D, B Adams L, Montana L, Kahn K, et al. The Association Between HIV-Related Stigma and the Uptake of HIV Testing and ART Among Older Adults in Rural South Africa: Findings from the HAALSI Cohort Study. AIDS Behav. 2024;28(3):1104–21. <u>https://doi.org/10.1007/s10461-023-04222-w</u> PMID: 38286975
- 59. Mahajan AP, Sayles JN, Patel VA, Remien RH, Sawires SR, Ortiz DJ, et al. Stigma in the HIV/AIDS epidemic: a review of the literature and recommendations for the way forward. AIDS. 2008;22 Suppl 2(Suppl 2):S67-79. <u>https://doi.org/10.1097/01.aids.0000327438.13291.62</u> PMID: <u>18641472</u>
- 60. World Health Organisation. SERVICE DELIVERY APPROACHES TO HIV TESTING AND COUN-SELLING (HTC): A STRATEGIC HTC PROGRAMME FRAMEWORK 2012. Available from: <u>https://</u> <u>mv.ecuo.org/download/services-delivery-approaches-to-hiv-testing-and-counselling-htc-a-strategichtc-programme-framework/</u>
- 61. UNAIDS. Evidence for eliminating HIV-related stigma and discrimination: Guidance for countries to implement effective programmes to eliminate HIV-related stigma and discrimination in six settings2020 09/01/2025 [cited 2025 09/01/2025]. Available from: <u>https://www.unaids.org/sites/default/</u>files/media_asset/eliminating-discrimination-guidance_en.pdf
- Levy ME, Phillips G 2nd, Magnus M, Kuo I, Beauchamp G, Emel L, et al. A Longitudinal Analysis of Treatment Optimism and HIV Acquisition and Transmission Risk Behaviors Among Black Men Who Have Sex with Men in HPTN 061. AIDS Behav. 2017;21(10):2958–72. <u>https://doi.org/10.1007/s10461-017-1756-z PMID: 28352984</u>
- 63. Kirakoya-Samadoulougou F, Jean K, Maheu-Giroux M. Uptake of HIV testing in Burkina Faso: an assessment of individual and community-level determinants. BMC Public Health. 2017;17(1):486. https://doi.org/10.1186/s12889-017-4417-2 PMID: 28532440